Author’s response to reviews

Title: Sodium valproate induced acute pancreatitis in a bipolar disorder patient: a case report

Authors:

Wanli Huang (pungpungli@outlook.com)

Xin Ren (monster_831@163.com)

Fang Shen (jek1998@126.com)

Baoping Xing (xbp2003@hotmail.com)

Version: 5 Date: 18 Nov 2019

Author’s response to reviews:

Dear Editors,
Thank you for your mail concerning our manuscript entitled “Sodium valproate induced acute pancreatitis in a bipolar disorder patient: a case report” (manuscript number: PHAT-D-18-00290R4). The main corrections in the paper and the responds to the editor’ comments are as followed:
1. In the ‘Funding’ section, changes have been made and the role of the funding body is addressed.
We appreciate for Editors’ warm work earnestly, and hope that the correction will meet with approval. Once again, thank you very much for the comments and suggestions.